Author’s response to reviews

Title: A Qualitative Exploration of Malaysian Cancer Patients' Perspectives on Cancer and its treatment

Authors:

Maryam Farooqui (maryamfaroogui.uitm@gmail.com)
Mohamed Azmi Hassali (azmihassali@gmail.com)
Aishah Knight Shatar (Aknight@amdi.usm.edu.my)
Asrul Akmal Shafie (aakmal@usm.my)
Tan Boon Seang (bseang@yahoo.com.sg)
Muhammad Aslam Farooqui (faroogi@acms.edu.my)

Version: 3 Date: 31 May 2011

Author’s response to reviews:

The Editor

BMC-series Journals

BioMed Central

London

Dear Sir/Madam,

Thank you very much for your comments on my manuscript MS: 5481827975157456. Below are my responses to the worthy concerns raised by the reviewers:

Reviewer 1: Wendy Lipworth

Comments: The authors still refer to their three major categories of analysis as "themes" (beliefs re causes of cancer, beliefs re effectiveness of conventional therapy, beliefs and experiences of adverse events). This is confusing as they then go on to describe their analysis of transcripts "for relevant content to identify emerging themes"

Response: Modified as suggested
Comments: The writing is much improved but could still do with a bit more editing.

Response: The manuscripts was revised for further editing

Thank you very much, I look forward for your kind response

Maryam Farooqui

Faculty of Pharmacy

Universiti Teknologi MARA (UiTM), Pulau Pinang

13500, Seberang Perai, Pulau Pinang

Malaysia